Title: Association between pulmonologists' tobacco use and their efforts in promoting smoking cessation during their routine clinical practice in Turkey: a cross-sectional study

Authors:

Pinar Pazarli Bostan (pinarpazarli@yahoo.com)
Canan Karaman Demir (drcanankaraman@yahoo.com)
Osman Elbek (osmanelbek@yahoo.com)
Sule Akcay (msuleakcay@gmail.com)

Version: 7
Date: 30 September 2015

Author’s response to reviews:

1. Title was modified.

2. In Line 70 and line 270: the parenthesis were deleted

In Line 273: "have further influence on physicians in promoting smoking cessation, independent from their", was changed in "have further influence on physicians in promoting smoking cessation, regardless of their".

3. The manuscript was edited by EDANZ.

4. We used the 1st paragraph of the Conclusion section in order to rephrase Abstract Conclusion.